Author's response to reviews

Title: Transient hypercortisolism and symptomatic hyperthyroidism associated to primary hyperparathyroidism in an elderly patient: case report and literature review

Authors:

Chiara Sabbadin (chiara.sabbadin.1@studenti.unipd.it)
Gabriella Donà (gabriella.dona@unipd.it)
Luciana Bordin (luciana.bordin@unipd.it)
Maurizio Iacobone (maurizio.iacobone@unipd.it)
Valentina Camozzi (valentina.camozzi@unipd.it)
Caterina Mian (caterina.mian@unipd.it)
Decio Armanini (decio.armanini@unipd.it)

Version: 5 Date: 4 January 2015

Author's response to reviews: see over
Dear Dr. Timothy Shipley
Executive editor of BMC Endocrine Disorders

The paper entitled: “TRANSIENT HYPERCORTISOLISM AND SYMPTOMATIC HYPERTHYROIDISM ASSOCIATED TO PRIMARY HYPERPARATHYROIDISM IN AN ELDERLY PATIENT: CASE REPORT AND LITERATURE REVIEW” has been revised and corrected according to the following Editor’s observations:

- "I have doubts though re the language. For example the authors use word "recognition" in the meaning of "diagnosis" as I understand (e.g. "In particular, the early recognition and treatment of a 40 transient post-operative thyrotoxicosis could improve recovery")."

- Requesting for Copy-Edit: We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Thank you for your annotations. The language of the manuscript has been fully reviewed and we have stressed the changes.
We hope that you will find this revised version of the manuscript suitable for the readers of BMC Endocrine Disorders.

Best regards,

Chiara Sabbadin

Corresponding author:
Department of Medicine-Endocrinology, University of Padua,
Via Ospedale 105, 35128 Padua, Italy
Telephone +39 049 8213023; Fax +39 049 657391
E-mail: chiara.sabbadin.1@studenti.unipd.it